Author's response to reviews

Title: Higher urine 1-hydroxy pyrene glucuronide (1-OHPG) is associated with tobacco smoke exposure and drinking mate in healthy subjects from Rio Grande do Sul, Brazil

Authors:

Renato B Fagundes (rfagundes@pro.viars.com.br)
Christian C Abnet (abnetc@mail.nih.gov)
Paul T Strickland (pstrickl@jhsph.edu)
Farin Kamangar (kamangaf@mail.nih.gov)
Mark J Roth (rothm@mail.nih.gov)
Philip R Taylor (taylorp@dcpcepn.nci.nih.gov)
Sanford M Dawsey (dawseys@dcpcepn.nci.nih.gov)

Version: 6 Date: 14 May 2006

Author's response to reviews:

1. Authors' contributions. Please include the following at the end of the declaration: "All authors read and approved the final manuscript".
   The declaration was added.

2. References. The reference list should contain all authors names, regardless of how many there are. The term et al should not be used. EndNote or Refman references. If your references are formatted using EndNote or Refman, please make any changes in the EndNote or Refman record, rather than editing the references in the manuscript manually.
   It was done.

3. Figures - Please do not use Roman numerals when numbering figures.
   Corrected